Author's response to reviews

Title: Seasonal variance of 25-(OH) vitamin D in the general population of Estonia, a Northern European country

Authors:

Mart Kull Jr (martkull@gmail.com)
Riina Kallikorm (riina.kallikorm@kliinikum.ee)
Anu Tamm (anu.tamm@kliinikum.ee)
Margus Lember (margus.lember@kliinikum.ee)

Version: 5 Date: 2 January 2009

Author's response to reviews:

1) The vertical guides were removed and cells merged over 2 rows unmerged.
2) 'Reference list' changed to 'References'.
3)'Figure title and legends' changed to 'Figure legends'.
4) Page 20 layout changed to portrait.
5) Removed blank page (page 21).